Author's response to reviews

Title: A randomized, double blind, crossover, placebo-controlled clinical trial to assess the effects of a mouthwash containing chlorine dioxide on oral malodor

Authors:

Kayoko Shinada (shinada.ohp@tmd.ac.jp)
Masayuki Ueno (ueno.ohp@tmd.ac.jp)
Chisato Mori (mori.ohp@tmd.ac.jp)
Sachiko Takehara (takehara.ohp@tmd.ac.jp)
Sayaka Yokoyama (yokoyama.ohp@tmd.ac.jp)
Yoko Kawaguchi (yoko.ohp@tmd.ac.jp)

Version: 2 Date: 11 April 2008

Author's response to reviews:

Dear Editors

I submitted our revised manuscript entitled A randomized, double blind, crossover, placebo-controlled clinical trial to assess the effect of a mouthwash containing chlorine dioxide on oral malodor which we should like considered for publication in Trials.

Please let me know of your decision at your earliest convenience. Thank you for your consideration.

Sincerely yours,

Kayoko Shinada DDS, PhD
Department of Oral Health Promotion, Graduate School,
Tokyo Medical and Dental University
1-5-45 Yushima, Bunkyo-ku, Tokyo 113-8549, Japn.
TEL: 81-3-5803-5476 FAX: 81-3-5803-0194
E-mail: shinada.ohp@tmd.ac.jp